Author’s response to reviews

Title: Maternal and Paternal Perinatal Depressive Symptoms Associate with 2- and 3-Year-Old Children’s Behaviour: Findings from the APrON Longitudinal Study

Authors:
Nicole Letourneau (nicole.letourneau@ucalgary.ca)
Brenda Leung (brenda.leung@uleth.ca)
Henry Ntanda (henry.ntanda@ucalgary.ca)
Deborah Dewey (dmdewey@ucalgary.ca)
Andrea Deane (andrea.deane@albertahealthservices.ca)
Gerald Giesbrecht (ggiesbre@ucalgary.ca)

Version: 2 Date: 01 Oct 2019

Author’s response to reviews:

Please refer to the summary of revisions document.